Author’s response to reviews

Title: Health care managers in negative media focus: a qualitative study of personification processes and their personal consequences

Authors:

Maria Wramsten Wilmar (maria.wramsten.wilmar@psy.gu.se)
Christian Jacobsson (christian.jacobsson@psy.gu.se)
Gunnar Jr Ahlborg (gunnar.ahlborg@vgregion.se)
Lotta Dellve (lotta.dellve@amm.gu.se)

Version: 8  Date: 13 December 2013

Author’s response to reviews:

Dear editors and referees,

thanks for the positive respons of our revisions. We have followed the two minor suggestions (1) a careful professional chack of language, (2) information about gender in figure 1.

Kindest regards